Author’s response to reviews

Title: A systematic review of effectiveness and safety of different regimens of levonorgestrel oral tablets for emergency contraception

Authors:

Mohammad Shohel (mohammadnsu@yahoo.com)
Mohammad M Rahman (dhakaiyamhfuz@yahoo.com)
Asif Zaman (asifzaman.nsu@gmail.com)
Mir MN Uddin (nasirmir5767@gmail.com)
MD M Al-Amin (bd_pharmacy@yahoo.com)
Hasan M Reza (rezami@hotmail.com)

Version: 3 Date: 9 August 2013

Author’s response to reviews:

1. title is changed as:

A systematic review of effectiveness and safety of different regimens of levonorgestrel oral tablets for emergency contraception

2. a line added in abstract in background section and that line is as below:

This study is a systematic review of the available evidence